Author’s response to reviews

Title: Acute Biliary Events During Anti-Tuberculosis Treatment: Hospital Case Series and Nationwide Cohort Study

Authors:

LIH-YU CHANG (pagasus2751@yahoo.com.tw)
Chih-Hsin Lee (chlee.tw@gmail.com)
Chia-Hao Chang (b89402012@ntu.edu.tw)
Ming-Chia Lee (ymkbaz60@gmail.com)
Meng-Rui Lee (sheepman1024@gmail.com)
Jann-Yuan Wang (jywang@ntu.edu.tw)
Li-Na Lee (linalee@ntu.edu.tw)

Version: 2 Date: 01 Dec 2017

Author’s response to reviews:

Responses to Reviewer #2

1. Results section, NHIRD Cohort (page 17, Line 58/59):

p value for DM is written 0.002, but in table 3 p value for DM is 0.001. Which one is correct? In addition authors can consider using asterisks in table 3, where p values are significant.

Answer: Thanks for reviewer’s careful review and we apologize for the typo. We have revised the p value in the text to be <0.001 (last sentence of the results section, page 17, Line 1/59) and modified Table 3 according to your suggestion. Thank you for the comment.

2. Authors need to thoroughly check the typos/language: for example:

Page 22, line 39, Please correct: usingrifamycin
Page 22, Line 55: Please correct: missingmild
Page 22, Line 58, Please correct: followingserum
Page 23, Line 39: Please correct: especiallyelderly
Answer: Thank you for the reminders. We apologize for the typos (missing one space between two words) and have corrected them in the revised manuscript. Thank you.